Author's response to reviews

Title: A population-based study of race-specific risk for placental abruption

Authors:

Tammy T Shen (shent@msnotes.wustl.edu)
Emily A DeFranco (defrancoe@wudosis.wustl.edu)
David M Stamilio (stamiliod@wudosis.wustl.edu)
Jen Jen Chang (jjchang@slu.edu)
Louis J Muglia (muglia_l@kids.wustl.edu)

Version: 5 Date: 1 September 2008

Author's response to reviews:

Dear Editorial Production Team,

We have made the following requested formatting changes to our manuscript:

1. We have removed the authors' qualifications from the manuscript and submission system.

2. We have provided full affiliation details in the order indicated, with linking to authors by superscript numbers.

3. We have inserted commas between the surname and forename initial as requested. This format does not seem to be the usual BMC Pregnancy and Childbirth format, but we have modified the references as asked.

Please let me know should you wish further changes. Thank you for your assistance.

Sincerely,

Louis Muglia